Author's response to reviews

Title: Recommended practices for computerized clinical decision support and knowledge management in community settings: A qualitative study

Authors:

Joan S Ash (ash@ohsu.edu)
Dean F Sittig (dean.f.sittig@uth.tmc.edu)
Kenneth Guappone (kenneth.guappone@providence.org)
Richard Dykstra (dykstrar@ohsu.edu)
Joshua Richardson (jor2032@med.cornell.edu)
Adam Wright (awright5@partners.org)
James Carpenter (james.carpenter@providence.org)
Carmit McMullen (carmit.mcmullen@kpchr.org)
Michael Shapiro (shapirom@ohsu.edu)
Arwen Bunce (buncea@ohsu.edu)
Blackford Middleton (bmiddleton1@partners.org)

Version: 4 Date: 25 January 2012

Author's response to reviews: see over
January 25, 2012

Dear Editor:

We are resubmitting our manuscript titled “Recommended Practices for Clinical Decision Support and Knowledge Management in Community Settings: A Qualitative Study” and appreciate your accepting it on principle pending your two requests.

1. Include the names of the places from which we received human subjects approval: these are now listed on pages 8 and 9.

2. Review the authors’ checklist for manuscript formatting to assure that our paper complies: we have done this.

In summary, we have made the revisions requested and we look forward to seeing the final publication.

Sincerely,

Joan S. Ash, Ph.D., M.B.A., M.L.S.
Professor and Vice-Chair